
-----------------------------------------------------------------------------
                Mauna Loa Solar Observatory Observer’s Log 
-----------------------------------------------------------------------------
        Wed Jun 17 16:53:01 GMT 1998
Year: 98  Doy: 168
Observer: koon
WEATHER COMMENT: Wed Jun 17 16:53:04 GMT 1998
Mostly cloudy with medium to thick cirrostratus, wind=5 mph from the SSE,
temp=52 F.
Wed Jun 17 16:56:26 GMT 1998    CHIP     Startup--Initializing new tape
Wed Jun 17 17:01:38 GMT 1998    CHIP     CHIP Start Patrol
PSPT COMMENT: Wed Jun 17 17:03:41 GMT 1998
Too cloudy for this.
MKIII COMMENT: Wed Jun 17 17:03:50 GMT 1998
Too cloudy for this.
PICS  COMMENT: Wed Jun 17 17:04:00 GMT 1998
Too cloudy for this.
WEATHER COMMENT: Wed Jun 17 17:42:58 GMT 1998
Clouds are too thick for spar to track.
CHIP COMMENT: Wed Jun 17 17:43:24 GMT 1998
Pausing due to clouds.
Wed Jun 17 17:43:49 GMT 1998    CHIP     CHIP End Patrol
CHIP COMMENT: Wed Jun 17 18:15:36 GMT 1998
Restarted Patrol, spar is tracking again.
art Patrol
MKIII COMMENT: Wed Jun 17 18:44:43 GMT 1998

***RECYCLING the last tape, H01852***

Wed Jun 17 19:01:43 GMT 1998    CHIP     Bias
Wed Jun 17 19:03:19 GMT 1998    CHIP     End Bias
Wed Jun 17 19:03:27 GMT 1998    CHIP     Water
Wed Jun 17 19:04:14 GMT 1998    CHIP     End Water
CHIP COMMENT: Wed Jun 17 19:32:37 GMT 1998
Paused due to clouds.
Wed Jun 17 19:34:44 GMT 1998    CHIP     CHIP End Patrol
COMMENT: Wed Jun 17 21:37:39 GMT 1998
I’ve been checking the new cmelst and the return messages that Eric got
on Monday.  Some of those addresses on the returned messages are very old,
like selvax.span, I changed that years ago.  I don’t understand why but
somehow the old addresses were used.
 There is still a problem getting our mail relayed beyond Boulder, Leonard
is working on it.
Wed Jun 17 21:53:18 GMT 1998    CHIP     ending tape
COMMENT: Wed Jun 17 22:01:46 GMT 1998

TAPE:
*****



CHIP: C00614
LOWL: L00524 in drive #0

Wed Jun 17 22:02:41 GMT 1998 
        MkIII   
 


